Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
May 26, 2010
Injector    Energy 56.62 MeV
· Injector needs optimization before Hall A is back on-line.

· (5/23) Cavity #8 of the quarter cryo module malfunctioned. The control module was replaced. 

· Since Hall B was the only hall taking beam for the last few days sometimes Hall B laser needed to be adjusted to compensate for drifts.  
·  (5/18) Injector Wien flipped for Hall A. Hall A Compton rates were very high after the Wien flip. Joe is going to revisit the injector.
· (5/12) The injector orbits were optimized to stabilize beam. 
Accelerator   Linac Energy 503.25 MeV
· The accelerator was running well and the optics stable.

· (5/25) BSY dump window re-pressurized. We should be able to run beam there without problems.

· (5/25) Zone 2L17 LCW leaked over the weekend. The leaking hose was replaced.
· (5/24) RSEP9A mismatched and was repaired.
· Overall the accelerator was running well with low trip rates. Main machine optics has been stable.

· After beam studies Ops recovers the accelerator very quickly.
· A lot of 4ch BPMs indicate ‘over saturation’ alarms. P. Francis has been investigating.

· (5/15) PSS dropped the injector to power permit for unknown reasons.

· (5/14) Injector BLM HV faulted. Ops cycled the power by unplugging it and putting it back.
· (5/13) Hall D construction work caused a power outage in East Arc during beam studies. Beam was restored three hours later.

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· Hall A stopped taking beam on May 20 due to its vacuum issues. 
· (5/18) After the injector Wien flip the Compton rates were high.

· Hall A was off more than seven shift due to the issues with its target chamber vacuum and spectrometer magnets. Friday evening the hall started to take beam.

· Arc1 sextupoles affect beam even the fields are set to zero. Make sure the trim cards be pulled after beam studies 
Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· (5/25) Hall B beam was not stable after yesterday’s beam study. Five hours was lost.

· No issues with Hall B beam. 

      Once in a while Hall B laser phase needed to be adjusted a degree or two to compensate

            for drifts since Hall A did not take beam.
· Hall B beam is good.
Hall C---QWeak from June 8 to August 15, 2010
· Hall C is planning to try beam on June 8.

· QWeak installation
